Author's response to reviews

Title: Analysis of open-water ultra-distance swimming performances in the FINA World Cup races in 5km, 10km and 25 km from 2000 to 2012

Authors:

Matthias A Zingg (zinadeuni@gmx.net)
Christoph A Rüst (christoph.ruest@bluewin.ch)
Thomas Rosemann (thomas.rosemann@usz.ch)
Romuald Lepers (romuald.lepers@u-bourgogne.fr)
Beat Knechtle (beat.knechtle@hispeed.ch)

Version: 2
Date: 23 December 2013

Author's response to reviews:

1. Title page: e-mail addresses are inserted
2. Ethics: All procedures used in the study met the ethical standards of the Swiss Academy of Medical Sciences [23] and were approved by the Institutional Review Board of Kanton St. Gallen, Switzerland with a waiver of the requirement for informed consent of the participants given the fact that the study involved the analysis of publicly available data.
3. Acknowledgements: no acknowledgements